Author’s response to reviews

Title: A survivin-responsive, conditionally replicating adenovirus induces potent cytocidal effects in adult T-cell leukemia/lymphoma

Authors:

Shinsuke Suzuki (suzuki2@m.kufm.kagoshima-u.ac.jp)
Hiroki Kofune (hkofune@kokando.co.jp)
Kimiharu Uozumi (uozumi9697kmc@gmail.com)
Makoto Yoshimitsu (myoshimi@m.kufm.kagoshima-u.ac.jp)
Naomichi Arima (nao@m2.kufm.kagoshima-u.ac.jp)
Kenji Ishitsuka (kenji-i@m.kufm.kagoshima-u.ac.jp)
Shin-ichi Ueno (ueno1@m.kufm.kagoshima-u.ac.jp)
Ken-ichiro Kosai (kosai@m2.kufm.kagoshima-u.ac.jp)

Version: 4  Date: 13 May 2019

Author’s response to reviews:

May 13, 2019

Linda Gummlich
The Editor
BMC Cancer

Dear Prof. Gummlich,

We thank you for careful reading our manuscript and for giving useful comments. In response to your comments, we have revised the manuscript entitled “A survivin-responsive, conditionally replicating adenovirus induces potent cytocidal effects in adult T-cell leukemia/lymphoma (ms no. BCAN-D-18-02765)”.


We look forward to a publication of our manuscript in BMC Cancer.

Sincerely,
Shinsuke Suzuki

Our responses to the Editor Comments are as follows:

Response for the Editor

Thank you for your suggestion.
We agree with you and have incorporated this suggestion throughout our paper.

1. We have re-written the sentences and phrases in our own words to ensure no overlap.

2. Because our manuscript does not contain any individual person’s data, we have stated “Not applicable” in The 'Consent for publication' section.